Author's response to reviews

Title: Association between the longest job and oral health: Japan Gerontological Evaluation Study project cross-sectional study

Authors:

Tatsuo Yamamoto (yamamoto.tatsuo@kdu.ac.jp)
Katsunori Kondo (kkondo@chiba-u.jp)
Jun Aida (aidaijun@m.tohoku.ac.jp)
Shinya Fuchida (fuchida@kdu.ac.jp)
Yukio Hirata (hirata@kdu.ac.jp)

Version: 4
Date: 10 October 2014

Author's response to reviews: see over
October 10, 2014

Dear Dr. Wael Sabbah:

I am submitting herewith our revised manuscript entitled, “Association between the longest job and oral health: Japan Gerontological Evaluation Study project cross-sectional study” for further consideration for publication in *BMC Oral Health*.

We wish to thank the editor for his careful critique of our manuscript. We have made every effort to implement his recommendations, and we feel that the paper has been improved as a result. The revised parts are highlighted in blue. We hope you will find the revised manuscript suitable for publication.

Sincerely yours,

Tatsuo Yamamoto
Department of Dental Sociology,
Graduate School of Dentistry,
Kanagawa Dental University,
82 Inaoka-cho, Yokosuka,
Kanagawa 238-8580, Japan
TEL & FAX: +81 46 822 8838
e-mail: yamamoto.tatsuo@kdu.ac.jp

**Response to Editor**
Consent statement. Please state in the Methods section whether written informed consent for participation in the study was obtained from participants or, where participants are children, a parent or guardian.

**Our response:** Thank you for your comments. We have added an explanation in the Methods section (page 8, lines 21 – page 9, line 1) as suggested.